
JUC, 21-4-2017, 15.00

present Des, Benito, Jonathan, Arpad, Harro via telephone

Des: just tried an (simulated)  e-test, failed as mark5 reservation system
failed. eBob to fix it. Most changes should be done, Now have to retry
disk-based correlation to see if that is not broken in the process.

Jonathan: fixed the problem with start on odd or even sample. The phases
in the 64 and 32 personalities now show the same trend, but noisier. Will
do more simulations. Weights were strange, this morning only zeros out of
UB. Restart, all fine again. Need to keep an eye on this.

Des: latest cpp sender has no slow-down. Harro: indeed, was just a hack.
Des: new version of control code should solve this with round robin.

Benito: still trying image with structure. Checking ms. Weights at 15? Same
data set, only differerent scans, more of them, and spectral averaging
done in UB, not afterwards. Will check this, might have to do with funny
state of UB? On Monday.


